Author's response to reviews

Title: Factors associated with HIV infection and transmissible behaviors among male injection drug users: a cross-sectional study in Long An, Vietnam

Authors:

Thu Minh T Tran (thuminh1105@yahoo.com)
Hien N Tran (nathiennihe@vnn.vn)
Hiroshi Yatsuya (h828@med.nagoya-u.ac.jp)
Nobuyuki Hamajima (nhamajim@med.nagoya-u.ac.jp)
Akio Nishimura (nishimur@med.nagoya-u.ac.jp)
Katsuki Ito (katsuki@med.nagoya-u.ac.jp)

Version: 2 Date: 9 February 2006

Author's response to reviews:

10 February 2006

Dr. Annabel Phillips
Assistant Editor
BMC-series Journals

Dear Dr. Phillips,

We would like to submit the revised manuscript including the name of the research ethics committee that approved the study, and also what form of consent had been obtained in the paragraph describing Population and sampling methods in the Methods section of the revised manuscript (P5, L21). We again wish it to be evaluated as a research paper for publication in "BMC Public Health" because the journal guarantees open access to all the readers in the world. The manuscript has been read and approved by all authors.

Sincerely yours,

Submitting author
Dr. Hiroshi Yatsuya
Department of Public Health
Nagoya University Graduate School of Medicine
65 Tsurumai-cho, Showa-ku, Nagoya 466-8550 Japan
fax: 81-52-744-2131
tel: 81-52-744-2128
e-mail: h828@med.nagoya-u.ac.jp